Author’s response to reviews

Title: Evaluating the feasibility of a nurse-led self-management support intervention for kidney transplant recipients: a pilot study

Authors:

Janet Been-Dahmen (j.m.j.been-dahmen@hr.nl)
Denise Beck (d.beck@erasmusmc.nl)
Mariëlle Peeters (m.a.c.peeters@hr.nl)
Heleen van der Stege (h.a.van.der.stege@hr.nl)
Mirjam Tielen (m.tielen@erasmusmc.nl)
Marleen van Buren (m.c.vanburen@erasmusmc.nl)
Erwin Ista (w.ista@erasmusmc.nl)
AnneLoes van Staa (a.van.staa@hr.nl)
Emma Massey (e.massey@erasmusmc.nl)

Version: 2 Date: 11 Jan 2019

Author’s response to reviews:

As requested, we have added the following sentence to page 7: All participants were being treated according to a standard protocol. Most newly transplanted patients start on a standard regimen of tacrolimus, mycophenolate mophetil acid (MMF), and prednisone for the first 6 months.

No other alterations were made.

We thank the reviewers and the editors for their time and consideration.